Author's response to reviews

Title: Impact of the Chinese new warning labels on knowledge, cigarette giving and quitting

Authors:

Yu Qin (jscdc_qinyu@sina.com)
Ming Wu (jswuming@vip.sina.com)
Xiaoqun Pan (panxq@yahoo.com.cn)
Quanyong Xiang (quanyongxiang@yahoo.com.cn)
Jianping Huang (nthjp@sohu.com)
Zenghui Gu (mbk@zjgcdc.cn)
Zumin Shi (zumin.shi@gmail.com)
Minghao Zhou (zmh@jscdc.cn)

Version: 2  Date: 24 May 2010

Author's response to reviews: see over
Dear Editor

Dr. Zumin Shi had looked over the manuscript and we have revised our manuscript. Dr. Shi is now working in a university of Australia, and he has been published several articles in English.

We have revised our title, from ‘The Chinese new warning labels of cigarette package was less effective on knowledge, cigarette giving and quitting------ A survey in Jiangsu Province’ to ‘Impact of the Chinese new warning labels on knowledge, cigarette giving and quitting’. See the title page.

We have added some results in the abstract part. See the abstract part.

We have added a statement of ethics approval in the methods part. See the methods part.

We have added a 'Competing interests' section between the Conclusions and Authors' contributions.

We also changed some grammar and spell error in the manuscript.

We look forward to your response.

Error! Reference source not found.

Yu Qin